Author’s response to reviews

Title: Epithelial Changes with Corneal Punctate Epitheliopathy in Type 2 Diabetes Mellitus and Their Correlation with Time to Healing

Authors:

Jinghao Qu (qujinghaochn@hotmail.com)
Li Li (eyelili@126.com)
Lei Tian (tianlei0131@163.com)
Xiaoyu Zhang (zhangxy199008@163.com)
Xuguang Sun (sunxg1955@163.com)
Ravi Thomas (rt@ravithomas.com)

Version: 2 Date: 12 Nov 2017

Author’s response to reviews:

Dear Editor,

Thank you for your letter and for the reviewers’ comments concerning our manuscript entitled “Epithelial Changes with Corneal Punctate Epitheliopathy in Type 2 Diabetes Mellitus and Their Correlation with Time to Healing” (MANUSCRIPT ID: BOPH-D-16-00467). We have studied their comments carefully and have made correction which we hope meet with their approval. Following was the list of actions and responses to reviewers.

List of Actions (LOA):

LOA1: The statement in Discussion section was revised to emphasize the coordination (Discussion section, line 214-255, page 7-8).

Reviewer reports:

Cirous Deghani (Reviewer 1):
The authors have appropriately addressed most of the comments raised by reviews. Although the revised version has now significantly been improved, but I still think that the discussion needs more input. For example, my first comment under Discussion has been addressed in the authors' response but there is no change in the manuscript. As well, please submit a clean version where all the changes are made, not in track changes. You have already addressed the changes in your response letter.

A: In accordance with reviewer’s request has been amended.